 Vc/20-Apr-04
Week 17 starting 19April 04

SM18 Operation Meeting (Tuesdays)

 Bench status  and tests for now & next few days : 

	Bench
	Name
	Status Mon Aft 19/4
	Temp

	TBB1
	 
	New Magnet ?
	300 K

	TBB2
	MBAR3066
	ICS 2 Connect Magnet (ICS)
	300 K

	TBC1
	MBAL3112
	Prep 5 Pumping + Cool Down
	Cooling

	TBC2
	SSS015
	New Magnet arrived
	300 K

	TBD1
	MBAL1077
	Prep 5 Pumping + Cool Down
	Cooling

	TBD2
	MBBL3129
	PT 10 Min. NRJ 11850 A
	1.9 K

	TBE1
	 
	new magnet ?
	300 K

	TBE2
	MBBL3073
	Prep 3 CDW1 HV
	300 K

	TBF1
	MBBL2032
	CR 1 Leak Test
	300 K

	TBF2
	MBBL3069
	CR 1 Leak Test
	 


Test Coordinator this week: Stephane ; MTM support on call  Maryline:  (16-3602).

9 Dipoles  Disconnected/removed  after tests last week    ( Sat Midnight to Sat midnight)  
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
· General Problems last week (gijs)
 Other General Issues
· Special Tests on 2032  -   F1 (Juan &  marco+ …. 
· D Cluster Commissioning of 600A & 150 A Power Supplies for future SSS setup on this cluster ( Guy + Giorgio)
· Keep D1 free for SSS Mech Preparation ( 6 hrs ? )   J.Halik 
	Date
	Bench
	Problem
	Type
	Application/soft
	Action/advise

	Tuesday 13_N
	TBB1
	Unable to launch Min Energy from TEMA. HF, LF application not starting


	AB/CO
	HF,LF
	Richard called. Off/On on all the crates. Then restart all. Then OK

	Wednesday 13_A
	TBE1
	SunMTA25 crashed. 
	AB/CO
	
	Richard called. He rebooted the sun

	Wednesday 13_N
	TBF1
	8th training quench: Slow PA.

Origin not known. 
	?
	
	Richard advised to re-launch the ramp. Then OK

	Thursday_15_M
	TBF1
	Fast de-excitation test. Cryo lost at 12000 A 
	ACR
	
	

	Thursday_15_A
	TBF1
	Application server not launched. Test could not be launched.
	
	
	

	Thursday_15_A
	TBF1
	CDAP: HV test not started because megger could not be put in remote mode.  The same for the other racks
	AB/CO
	
	Pedro contacted by Richard.Faulty synchronisation between Megger and program/

Corrected.

	Friday_16_A
	TBD1
	MBA 1077. Scan validation error: magnet did not exist. 
	ICS
	MTF
	Richard called. He advised to check in MTF. MTF says that it is a type A magnet, confirmed with decapole measurements in WP1. Not correct labelling the anticryostat.

	Monday_19_M
	TBD2
	Training quench. The current ramped up to 9000 A and down the up again.
	AB/CO
	TEMA
	Measurements stopped. Maryline, Giorgio informed. They found a bug in labview. Corrected by Pedro et Alessandro.

	Monday_19_M
	TBE2
	Problem with HWREC, file snot generated.
	MTM
	HWREC
	Maryline informed. She restarted the sun and then OK.








